Author’s response to reviews

Title: Risk variants of the α-synuclein locus and REM sleep behavior disorder in Parkinson's disease: a genetic association study

Authors:

Kari Anne Bjørnarå (kari.anne.bjornara@vestreviken.no)
Lasse Pihlstrøm (lasse.pihlstrom@medisin.uio.no)
Espen Dietrichs (espen.dietrichs@medisin.uio.no)
Mathias Toft (mathias.toft@ous-hf.no)

Version: 2 Date: 05 Feb 2018

Author’s response to reviews:

To

Chin-Hsien Lin
Editor of BMC Neurology

Date: 5.02.2018

Case Worker:
Telephone:
Our reference:
Your reference:
Clinic/Department:

Subject: Manuscript # NURL-D-17-00535 “Risk variants of the α-synuclein locus and REM sleep behavior disorder in Parkinson's disease: a genetic association study”

Dear Editor,

We are very grateful that you have found our manuscript acceptable for publishing in BMC Neurology.

We have now uploaded the final manuscript for production.

Sincerely,
Kari Anne Bjørnarå, MD, PhD
Department of Neurology
Drammen Hospital
Vestre Viken Hospital Trust

Tel: +47 92668493
E-mail: kari.anne.bjornara@vestreviken.no